Author’s response to reviews

Title: What works in falls prevention in Asia: A systematic review and Meta-analysis of randomized controlled trials.

Authors:

Keith Hill (keith.hill@curtin.edu.au)
Plaiwan Suttanon (plaiwan2277@gmail.com)
Sang-I Lin (lin31@mail.ncku.edu.tw)
William Tsang (william.tsang@polyu.edu.hk)
Asmidawati Ashari (asmidawati@upm.edu.my)
Dr Tengku Aizan Hamid (tengkuaizan06@gmail.com)
Kaela Farrier (kaela.farrier@curtin.edu.au)
Elissa Burton (E.Burton@curtin.edu.au)

Version: 1 Date: 22 Nov 2017

Author’s response to reviews:

see attached response letter